The Prevention of Delirium system of care for older patients admitted to hospital for emergency care: the POD research programme including feasibility RCT

John Young,¹ John Green,^{1*} Mary Godfrey,¹ Jane Smith,¹ Francine Cheater,² Claire Hulme,³ Michelle Collinson,⁴ Suzanne Hartley,⁴ Shamaila Anwar,⁵ Marie Fletcher,⁴ Gillian Santorelli,⁶ David Meads,⁷ Keith Hurst,⁸ Najma Siddiqi,⁹ Dawn Brooker,¹⁰ Elizabeth Teale,¹ Alex Brown,¹¹ Anne Forster,¹ Amanda Farrin⁴ and Sharon Inouye¹²

- ¹Academic Unit for Ageing and Stroke Research, University of Leeds, Leeds, UK
 ²School of Health Sciences, University of East Anglia, Norwich, UK
 ³College of Medicine and Health, University of Exeter, Exeter, UK
 ⁴Clinical Trials Research Unit, University of Leeds, Leeds, UK
 ⁵National Institute for Health Research Clinical Research Network, Huddersfield, UK
 ⁶Born in Bradford, Bradford Institute for Health Research, Bradford, UK
 ⁷Academic Unit of Health Economics, University of Leeds, Leeds, UK
 ⁸Hurst Research Ltd, King's Lynn, UK
 ⁹Department of Health Sciences, University of York, Hull York Medical School, York, UK
- ¹⁰Association for Dementia Studies, University of Worcester, Worcester, UK
- ¹¹Elderly and Intermediate Care Service, Bradford Teaching Hospitals NHS Foundation Trust, Bradford, UK
- ¹²Harvard Medical School, Beth Israel Deaconess Medical Center, Marcus Institute for Aging Research, Boston, MA, USA

*Corresponding author john.green@bthft.nhs.uk

Declared competing interests of authors: Claire Hulme was a member of the National Institute for Health Research (NIHR) Health Technology Assessment (HTA) Commissioning Board (2013–17). David Meads reports previous membership of the HTA European Economic and Social Committee (EESC) Methods Group (2014–17) and the HTA EESC Panel (2013–17). Elizabeth Teale reports personal fees from MA Healthcare Conferences (London, UK), outside the submitted work. Anne Forster reports previous membership of the Health Services and Delivery Research Researcher-Led Panel (2016–18) and grants from the NIHR Programme Grants for Applied Research programme (RP-PG-1210-12017, RP-PG-0615-20019 and RP-PG-0611-20010), outside the submitted work. Amanda Farrin reports previous membership of the HTA Antimicrobial Resistance Themed Call Board (2013–14), the HTA Efficient Study Design Board (2014), the HTA Flu Themed Call Board (2009–11), the HTA Obesity Themed Call Board (2008–10), the HTA Pandemic Influenza Board (2009 and 2011),

the HTA Primary Care Themed Call Board (2013–14), the HTA Surgery Themed Call Board (2012–13), the Rapid Trials and Add-on Studies Board (2012–16), the HTA Funding Committee Policy Group (2014–18) and the HTA Clinical Evaluation and Trials Committee (2014–18).

Published March 2021 DOI: 10.3310/pgfar09040

Plain English summary

POD research programme including feasibility RCT Programme Grants for Applied Research 2021; Vol. 9: No. 4 DOI: 10.3310/pgfar09040

NIHR Journals Library www.journalslibrary.nihr.ac.uk

Plain English summary

Delirium (sometimes called acute confusion) is a serious condition in which there is a sudden change in a person's mental state. It affects about one-third of older people admitted to hospital; it can be very distressing and many people do not fully recover.

A system of care called the Hospital Elder Life Program was developed in the USA to reduce delirium occurrence in older patients admitted to hospital. We wanted to see if the Hospital Elder Life Program could work in NHS hospitals.

We found that the Hospital Elder Life Program was challenging to use in the NHS because of the extra resources needed. We therefore developed the Prevention of Delirium system of care, which could be used without additional staff. Of the six wards recruited, we tested the Prevention of Delirium system of care in five: it was found to be acceptable to patients, carers, staff and volunteers in most wards, and did not make extra work for nurses. We improved the Prevention of Delirium system of care and then undertook a further study in 16 wards in eight hospitals to see if the Prevention of Delirium system of care had the potential to reduce delirium, if it was likely to be cost-effective and if a larger study would be practicable.

We were able to recruit 713 patients (the target was 720) and successfully tested 712 (99.9%) of them at least once for delirium. We managed to follow up 400 (56.1%) patients for 3 months following their recruitment to the trial. Most wards managed to use the Prevention of Delirium system of care to a reasonable extent. The Prevention of Delirium system of care reduced delirium occurrence to a similar extent as that of other studies and may represent value for money, although a larger study is needed to be sure of these findings. Overall, the findings indicate that a larger study would be practicable and would be likely to provide more accurate results to decide if the Prevention of Delirium system of care is an effective system of care for general use in the NHS.

© Queen's Printer and Controller of HMSO 2021. This work was produced by Young *et al.* under the terms of a commissioning contract issued by the Secretary of State for Health and Social Care. This issue may be freely reproduced for the purposes of private research and study and extracts (or indeed, the full report) may be included in professional journals provided that suitable acknowledgement is made and the reproduction is not associated with any form of advertising. Applications for commercial reproduction should be addressed to: NIHR Journals Library, National Institute for Health Research, Evaluation, Trials and Studies Coordinating Centre, Alpha House, University of Southampton Science Park, Southampton SO16 7NS, UK.

Programme Grants for Applied Research

ISSN 2050-4322 (Print)

ISSN 2050-4330 (Online)

This journal is a member of and subscribes to the principles of the Committee on Publication Ethics (COPE) (www.publicationethics.org/).

Editorial contact: journals.library@nihr.ac.uk

The full PGfAR archive is freely available to view online at www.journalslibrary.nihr.ac.uk/pgfar. Print-on-demand copies can be purchased from the report pages of the NIHR Journals Library website: www.journalslibrary.nihr.ac.uk

Criteria for inclusion in the Programme Grants for Applied Research journal

Reports are published in *Programme Grants for Applied Research* (PGfAR) if (1) they have resulted from work for the PGfAR programme, and (2) they are of a sufficiently high scientific quality as assessed by the reviewers and editors.

Programme Grants for Applied Research programme

The Programme Grants for Applied Research (PGfAR) programme, part of the National Institute for Health Research (NIHR), was established in 2006 to fund collaborative, multidisciplinary programmes of applied research to solve health and social care challenges. Findings are expected to provide evidence that lead to clear and identifiable patient benefits, in the relatively near future.

PGfAR is researcher led and does not specify topics for research; however, the research must be in an area of priority or need for the NHS and the social care sector of the Department of Health and Social Care, with particular emphasis on health and social care areas that cause significant burden, where other research funders may not be focused, or where insufficient funding is available.

The programme is managed by the NIHR Central Commissioning Facility (CCF) with strategic input from the Programme Director. For more information about the PGfAR programme please visit the website: https://www.nihr.ac.uk/explore-nihr/funding-programmes/programme-grants-for-applied-research.htm

This report

The research reported in this issue of the journal was funded by PGfAR as project number RP-PG-0108-10037. The contractual start date was in December 2009. The final report began editorial review in October 2019 and was accepted for publication in October 2020. As the funder, the PGfAR programme agreed the research questions and study designs in advance with the investigators. The authors have been wholly responsible for all data collection, analysis and interpretation, and for writing up their work. The PGfAR editors and production house have tried to ensure the accuracy of the authors' report and would like to thank the reviewers for their constructive comments on the final report document. However, they do not accept liability for damages or losses arising from material published in this report.

This report presents independent research funded by the National Institute for Health Research (NIHR). The views and opinions expressed by authors in this publication are those of the authors and do not necessarily reflect those of the NHS, the NIHR, CCF, NETSCC, PGfAR or the Department of Health and Social Care. If there are verbatim quotations included in this publication the views and opinions expressed by the interviewees are those of the interviewees and do not necessarily reflect those of the authors, those of the NHS, the NIHR, NETSCC, the PGfAR programme or the Department of Health and Social Care.

© Queen's Printer and Controller of HMSO 2021. This work was produced by Young *et al.* under the terms of a commissioning contract issued by the Secretary of State for Health and Social Care. This issue may be freely reproduced for the purposes of private research and study and extracts (or indeed, the full report) may be included in professional journals provided that suitable acknowledgement is made and the reproduction is not associated with any form of advertising. Applications for commercial reproduction should be addressed to: NIHR Journals Library, National Institute for Health Research, Evaluation, Trials and Studies Coordinating Centre, Alpha House, University of Southampton Science Park, Southampton SO16 7NS, UK.

Published by the NIHR Journals Library (www.journalslibrary.nihr.ac.uk), produced by Prepress Projects Ltd, Perth, Scotland (www.prepress-projects.co.uk).

NIHR Journals Library Editor-in-Chief

Professor Ken Stein Professor of Public Health, University of Exeter Medical School, UK

NIHR Journals Library Editors

Professor John Powell Chair of HTA and EME Editorial Board and Editor-in-Chief of HTA and EME journals. Consultant Clinical Adviser, National Institute for Health and Care Excellence (NICE), UK, and Professor of Digital Health Care, Nuffield Department of Primary Care Health Sciences, University of Oxford, UK

Professor Andrée Le May Chair of NIHR Journals Library Editorial Group (HS&DR, PGfAR, PHR journals) and Editor-in-Chief of HS&DR, PGfAR, PHR journals

Professor Matthias Beck Professor of Management, Cork University Business School, Department of Management and Marketing, University College Cork, Ireland

Dr Tessa Crilly Director, Crystal Blue Consulting Ltd, UK

Dr Eugenia Cronin Senior Scientific Advisor, Wessex Institute, UK

Dr Peter Davidson Consultant Advisor, Wessex Institute, University of Southampton, UK

Ms Tara Lamont Senior Scientific Adviser (Evidence Use), Wessex Institute, University of Southampton, UK

Dr Catriona McDaid Senior Research Fellow, York Trials Unit, Department of Health Sciences, University of York, UK

Professor William McGuire Professor of Child Health, Hull York Medical School, University of York, UK

Professor Geoffrey Meads Emeritus Professor of Wellbeing Research, University of Winchester, UK

Professor James Raftery Professor of Health Technology Assessment, Wessex Institute, Faculty of Medicine, University of Southampton, UK

Dr Rob Riemsma Reviews Manager, Kleijnen Systematic Reviews Ltd, UK

Professor Helen Roberts Professor of Child Health Research, UCL Great Ormond Street Institute of Child Health, UK

Professor Jonathan Ross Professor of Sexual Health and HIV, University Hospital Birmingham, UK

Professor Helen Snooks Professor of Health Services Research, Institute of Life Science, College of Medicine, Swansea University, UK

Professor Ken Stein Professor of Public Health, University of Exeter Medical School, UK

Professor Jim Thornton Professor of Obstetrics and Gynaecology, Faculty of Medicine and Health Sciences, University of Nottingham, UK

Please visit the website for a list of editors: www.journalslibrary.nihr.ac.uk/about/editors

Editorial contact: journals.library@nihr.ac.uk